Author's response to reviews

Title: Acute presentation of vasospastic angina induced by oral capecitabine: a case report.

Authors:

Christos Gkolias (kcharal@med.duth.gr)
Georgios Dimitriadis (georgiosdim@yahoo.gr)
Dimokritos Dimitriadis (dimitriadis.dimokritos@gmail.com)
Christos Graidis (c.graidis65@gmail.com)
Pavlos Tsinivizof (pavlost1980@gmail.com)
Afroditi Tsiakou (tsiakou1981@hotmail.com)
Theodosis Bitsis (theooo8@yahoo.com)
Konstantinos Charalabopoulos (christos_gkolas@yahoo.com)

Version: 4
Date: 1 November 2013

Author's response to reviews: see over
Dear Editor,

I hereby submit you a revised version of the article entitled "Acute presentation of vasospastic angina induced by oral capecitabine: a case report".

(MS: 1139669541007837)

All reviewer(s)' comments were taken into account and point to point answers are included at the bottom of this letter.

Please do not hesitate to contact me if necessary,
Thank you in advance,

Sincerely yours,

Prof. K.A. Charalabopoulos MD, PhD

Department of Physiology, Clinical Unit, Medical Faculty
Democritus University of Thrace
Panepistimioupoli, Dragana
68100, Alexandroupolis, Greece
Tel: 003 25510-30538
Fax: 003 25510-30538
E-mail: christos_gkoliyas@yahoo.com

POINT TO POINT CHANGES

IN GENERAL

All reviewer(s)' suggestions were taken into account.

IN PARTICULAR

1. English language was significantly improved.

2. The legends of the figures 2 and 3 were enriched and say what the figures show. Additionally, the legend mention the changes to note ("Electrocardiogram obtained during the anginal episode in the Coronary Care Unit. Sinus bradycardia (50bpm) with diffuse ST-segment elevation (arrows) in the anterior (V3-V6) and inferolateral leads (I,II,III,aVL,aVF) and peaked T-waves (dots) at the same leads are suggestive of transmural ischemia"), ("Electrocardiogram 20 minutes after the pain resolved. Arrows show progressive recovery of ventricular repolarization abnormalities")

3. Marks (arrows and dots) were added to figures 2 and 3 pointing to the relevant ECG changes.